Author's response to reviews

Title: Associations between neighbourhood walkability and daily steps in adults: A systematic review and meta-analysis

Authors:

Samantha Hajna (samantha.hajna@mail.mcgill.ca)
Nancy A Ross (nancy.ross@mcgill.ca)
Anne-Sophie Brazeau (anne-sophie.brazeau@mcgill.ca)
Patrick Bélisle (patrick.belisle@clinepi.mcgill.ca)
Lawrence Joseph (lawrence.joseph@mcgill.ca)
Kaberi Dasgupta (kaberi.dasgupta@mcgill.ca)

Version: 4
Date: 17 July 2015

Author's response to reviews: see over
July 16, 2015

Re: Manuscript 3031316051647913

Dear Dr. Melody Ding:

As you requested, we now only use initials in the Authors’ Contribution section and have placed the Competing Interests statement into a separate section with its own heading.

We trust that you will find our revised submission suitable for publication in BMC Public Health.

Thank you again for your feedback. We look forward to hearing from you.

Sincerely,

[Signature]

Kaberi Dasgupta, MD, MSc, FRCP(C)
Associate Professor of Medicine
Department of Medicine
McGill University